Author’s response to reviews

Title: Assessing the maturity of the healthcare system for integrated care: testing measurement properties of the SCIROCCO tool

Authors:

Liset Grooten (fennechien.grooten@vub.be)
Hubertus Vrijhoef (bert.vrijhoef@panaxea.eu)
Stefano Calciolari (stefano.calciolari@usi.ch)
Laura Ortiz (laura.guadalupe.gonzalez.ortiz@usi.ch)
Marcela Janečková (marcela.janeckova@gmail.com)
Mirella Minkman (M.Minkman@tias.edu)
Dirk Devroey (Dirk.Devroey@vub.be)

Version: 2 Date: 04 Mar 2019

Author’s response to reviews:

Dear Editor,

Thank you for your time reviewing the revised manuscript. Please find below our response to the request for some minor revisions in the revised manuscript.

1. In the Ethics approval and consent to participate section of your manuscript, please provide clarifications as to why consent was waived for this study.

Our response: We included the explanation on page 9 under the section "Ethics approval and consent to participate".
2. We note that the initials DV have been listed in the author's contributions section of your manuscript. However, these initials do not correlate with any of the listed authors. Please clarify whether this is a typing error or if this is an author that has not been listed. If you choose to change your author list you will need to fill out a change in authorship form and send it by email to the Editorial office to be approved by the Editor. The form can be found here: https://resource-cms.springernature.com/springer-cms/rest/v1/content/7454878/data/v5

Our response: This was indeed a typing error. We change the DV to DD (which refers to the author Dirk Devroey).

3. Please replace "SPSS" with "IBM SPSS Statistics".

Our response: We replaced the SPSS with "IBM SPSS Statistics".

Many thanks.

Kind regards,

Liset